# Multi-Aspect Conditioning for Diffusion-Based Music Synthesis: Enhancing Realism and Acoustic Control

Ben Maman<sup>(D)</sup>, Johannes Zeitler<sup>(D)</sup>, Meinard Müller<sup>(D)</sup>, Fellow, IEEE, and Amit H. Bermano<sup>(D)</sup>

Abstract-Music synthesis aims to generate audio from symbolic music representations, traditionally using techniques like concatenative synthesis and physical modeling. These methods offer good control but often lack expressiveness and realism in timbre. Recent advancements in diffusion-based models have enhanced the realism of synthesized audio, yet these models struggle with precise control over aspects like acoustics and timbre and are limited by the availability of high-quality annotated training data. In this paper, we introduce an advanced diffusion-based framework for music synthesis that further improves realism and introduces control through multi-aspect conditioning. This allows the synthesis from symbolic representations to accurately replicate specific performance and acoustic conditions. To address the need for precise multi-instrument target annotations, we propose using MIDIaligned scores and automatic multi-instrument transcription based on neural networks. These methods effectively train our diffusion model with authentic audio, enhancing realism and capturing subtle nuances in performance and acoustics. As a second major contribution, we adopt conditioning techniques to gain control over multiple aspects, including score-related aspects like notes and instrumentation, as well as version-related aspects like performance and acoustics. This multi-aspect conditioning restores control over the music generation process, leading to greater fidelity in achieving the desired acoustic and stylistic outcomes. Finally, we validate our model's efficacy through systematic experiments, including qualitative listening tests and quantitative evaluation using Fréchet Audio Distance to assess version similarity, confirming the model's ability to generate realistic and expressive music, with acoustic control. Supporting evaluations and comparisons are detailed on our website (benadar293.github.io/multi-aspect-conditioning).

#### Index Terms-Multi-Instrument synthesis, diffusion.

Received 30 June 2024; revised 22 October 2024; accepted 12 November 2024. Date of publication 28 November 2024; date of current version 24 January 2025. This work was supported in part by Len Blavatnik and the Blavatnik Family Foundation, in part by the ISF under Grant 1337/22, and in part by the Deutsche Forschungsgemeinschaft (DFG, German Research Foundation) under Grant 500643750 (MU 2686/15-1). The International Audio Laboratories Erlangen are a joint institution of the Friedrich-Alexander-Universität Erlangen-Nürnberg (FAU) and the Fraunhofer Institute for Integrated Circuits IIS. The associate editor coordinating the review of this article and approving it for publication was Prof. Juhan Nam. (*Corresponding author: Ben Manan.*)

Ben Maman, Johannes Zeitler, and Meinard Müller are with the International Audio Laboratories Erlangen, 91058 Erlangen, Germany (e-mail: ben.maman@ audiolabs-erlangen.de; johannes.zeitler@audiolabs-erlangen.de; meinard. mueller@audiolabs-erlangen.de).

Amit H. Bermano is with the Tel Aviv University, Tel Aviv-Yafo 6997801, Israel (e-mail: amberman@tauex.tau.ac.il).

This article has supplementary downloadable material available at https://doi.org/10.1109/TASLP.2024.3507553, provided by the authors. Digital Object Identifier 10.1109/TASLP.2024.3507553

#### I. INTRODUCTION

USIC synthesis, the process of generating audio from WI symbolic music representations, is an important and long-studied area of research with applications in music creation and production. Traditional signal processing approaches, employing techniques such as concatenative synthesis [1], [2], [3], [4] and physical modeling [5], [6], have been used to produce high-quality audio for specific instruments. These methods provide explicit control over the audio output, thanks to note-wise rendering and the manipulation of pre-recorded samples. Even though these methods offer a variety of timbres, they still restrict flexibility and often lack expressiveness and realism, especially in terms of timbre and acoustic conditions. For example, replicating the unique sound of a specific orchestra or instrument poses significant challenges, such as producing audio that resembles the Berlin Philharmonic Orchestra or emulating the guitar sound from a 1975 recording by Segovia.

Recent advancements in data-driven generative modeling have addressed some of these limitations by enabling the inference of semantic aspects from example data [7], [8], [9], [10], [11]. In particular, Denoising Diffusion Probabilistic Models (DDPMs) have been extremely successful in rendering realistic images and learning styles from example images [11], [12], [13], [14]. Similarly, such models have been used to learn sounds from examples, significantly improving the expressiveness and realism of synthesized audio, including speech [15], [16] and music [17], [18].

However, these data-driven models require large datasets with detailed annotations of notes and instruments, confining them to specific curated datasets or necessitating the use of lower-quality synthesized data. For example, the DDPM-based approach described by Hawthorne et al. [17] uses data based on concatenative synthesis, compromising audio quality and resulting in flat, less realistic audio.

Another problem is that these approaches often face challenges in precisely controlling aspects such as acoustics and timbre. In particular, these data-driven generative methods may encounter problems such as instrument drift, where the same instrument is not rendered coherently in different parts of the generated audio.

In this paper, building upon our previous work [21], we introduce an advanced diffusion-based framework for music synthesis that incorporates multi-aspect conditioning on notes,

<sup>© 2024</sup> The Authors. This work is licensed under a Creative Commons Attribution 4.0 License. For more information, see https://creativecommons.org/licenses/by/4.0/



Fig. 1. Schematic overview of the proposed approach for diffusion-based music synthesis using multi-aspect conditioning. Our model generates audio conditioned both on the *score*, and on the *version*. The score contains the notes, instruments and timing, providing the musical content. The version corresponds to the specific acoustic and performance aspects, including timbre, rocording environment, and style. Version conditioning provides control and flexibility, enabling to generate different timbres of the same instrument (e.g., different types of guitar), or different room acoustics.

instruments, performance styles, and acoustic details to significantly enhance realism and control. Consequently, our method facilitates the synthesis of music signals from symbolic representations, accurately replicating specific performance and acoustic conditions as provided in example recordings. For instance, our model is capable of reproducing the guitar sound from a 1975 recording of Segovia playing Albéniz's "Capriccio Catalán", now applied to another piece, such as Jobim's "Felicidad." Furthermore, even in more complex musical settings, our model can transfer certain timbre and acoustic conditions from orchestral recordings, such as Karajan's 1962 recordings of Beethoven's symphonies with the Berlin Philharmonic, to new musical works presented in symbolic format. To the best of our knowledge, our work is the first to address such challenges in a multi-instrument setting.

The main contributions of this paper, which substantially extends our initial study [21], can be summarized as follows. First, to meet the demand for precisely annotated multi-instrument symbolic data, we propose two approaches: one using MIDIaligned symbolic scores as in [21], and the other employing recent multi-instrument transcription methods [22]. We demonstrate that in both cases, the quality of the target data is sufficient to leverage uncurated multi-instrument real-world audio for training our diffusion model. This use of authentic training data enables our model to capture genuine musical performance characteristics, including subtle nuances in timbre and acoustics.

As a second main contribution, we expand the capabilities of diffusion models by integrating control over multiple musical aspects, including score-related aspects such as notes and instrumentation, as well as version-related aspects such as acoustics and performance. Specifically, we demonstrate how version control can be achieved using conditioning techniques based on Feature-wise Linear Modulation (FiLM) layers [23]. Our multi-aspect conditioning approach allows us to train a single model on massive amounts of uncurated multi-instrument performances with diverse instrumentation, including symphonic orchestras, chamber orchestras, church organs, harpsichords, violins, guitars, and more. Furthermore, we show how version conditioning restores control over the music generation process, leading to greater fidelity in achieving the desired acoustic and stylistic outcomes. For an overview of our DDPM-based model, see Fig. 2.

As a third contribution, we validate the efficacy of our model through rigorous evaluation methods. These include listening tests and the application of the Fréchet Audio Distance [24] to assess both realism and version similarity, which confirms the model's ability to generate realistic and expressive audio. These evaluations are complemented by a wide range of synthesized music examples and comparisons with prior methods, all accessible on our freely available website.<sup>1</sup>

The remainder of this paper is organized as follows. Section II discusses related work, focusing primarily on music synthesis. Section III introduces the computational pipelines of our diffusion-based approach to music synthesis. In Section IV, we present our main technical contribution, the multi-aspect conditioning framework, which includes our extensions of score conditioning using alignment and transcription techniques, with our proposed version conditioning. In Section V we present our proposed technique for consistent and smooth segmentwise generation. Section VI details the evaluation criteria and presents experiments, including qualitative listening tests and quantitative evaluations. Finally, Section VII concludes the paper and discusses future work, highlighting the potential of version conditioning for automated instrumentation achievable with pitch-only input.

## II. RELATED WORK

Audio synthesis in current literature can be done autoregressively, where models directly construct a waveform sample-by-sample [25], [31], [32]. Another approach, which we take, operates in the spectral domain. This requires a subsequent step to convert the generated spectral representation (whether the short-time Fourier transform or mel-spectrogram) into a waveform, but it is computationally more efficient.

For a data-driven approach, large amounts of labeled data are required, i.e., paired datasets of audio recordings and their corresponding time-aligned transcriptions. While such datasets exist for instruments such as the piano thanks to special equipment (e.g., Disklavier), this is not the case for other instruments. Thus, previous works mainly focus on generating piano performances, monophonic (single-voice) music (which is easier to label), or music produced by a concatenative synthesizer [17], [27], which is trivially supervised.

Table I provides a summary of existing methods for score-tomusic synthesis. Wang and Yang [26] use a U-Net to synthesize



Fig. 2. Overview of our proposed diffusion-based synthesis model. Our model learns to denoise mel-spectrograms conditioned on score-based and version-based information. Version conditioning (determining the acoustics, specific timbre, recording environment and style) is done through FiLM layers at each block, which can be applied to either a T5 transformer or a U-Net. The version condition is represented by the performance ID, and is inserted at each layer by concatenation with the diffusion timestep. The score condition is provided as additional input, in the form of a MIDI-like piano-roll representation. For sampling consistent segments with smooth transitions, we use an overlapped generation technique borrowed from Computer Vision [19], [20].

#### TABLE I Overview of Previous Work Indicating Ability to Render Multiple Instruments Simultaneously (Multi), Version Control (Version), Generating Orchestral Symphonies (Symph.), Data Size, and Ratio of Real Vs. Synthetic Data Used for Training (Real%)

	Multi	Version	Symph.	Data	Real%
[25]	X	<ul> <li>Image: A set of the set of the</li></ul>	X	$\sim 140 \text{H}$	100%
[26]	×	×	×	$\sim 1 \text{H}$	100%
[27]	×	×	×	$\sim 1 \text{H}$	0%
[28]	×	×	×	$\sim 1 \text{H}$	100%
[29]	×	×	×	$\sim 3H$	100%
[17]	<ul> <li>Image: A set of the set of the</li></ul>	×	×	$\sim 1500 \text{H}$	$\sim 2\%$
[30]	X	X	X	$\sim 93H$	$\sim 3\%$
Ours	<ul> <li>Image: A set of the set of the</li></ul>	<ul> <li>Image: A set of the set of the</li></ul>	<ul> <li>Image: A set of the set of the</li></ul>	$\sim 58H$	100%

solo violin, cello, and flute performances, requiring a separate model for each instrument. It is trained with a spectral reconstruction loss, without diffusion. Similarly, Dong et al. [28] use a Transformer architecture to synthesize solo violin or piano. Both works produce only monophonic and single-instrument music (i.e., only a single note or single instrument is synthesized at any given time).

Wu et al. [29] learn a parametric model of a musical performance, synthesizing from performance controls such as intensity, vibrato, etc. Although promising, a central drawback is that this work only operates on monophonic single-instrument data, similar to former works—due to the higher complexity of polyphonic music, and lack of high-quality polyphonic training data.

In multi-instrument synthesis, Hawthorne et al. [17] use a T5 Transformer-based diffusion model. While this method is promising and produces high-fidelity audio, it has limitations: It does not have control over the version, acoustics, and style (e.g., specific type of organ when several exist, or recording environment), and produces less realistic sound, due to lack of real annotated data, as demonstrated on our project page.

Kim et al. [30] use a diffusion-based approach following Hawthorne et al. [17], with a down-scaled T5 model, for guitar synthesis. The work is limited to guitar alone, and is trained mainly on synthetic data due to the aforementioned lack of real annotated data.

Other works exist for generating audio conditioned on text prompts [33], [34], however, in this work we focus on score-conditioned music synthesis.

## III. DIFFUSION-BASED MUSIC SYNTHESIS

An overview of our method is depicted in Fig. 2. We seek to enhance the generation quality and control of an off-theshelf diffusion-based music generator using a collection of real multi-instrument performances with corresponding scores and version information. Hence, we start from a dataset  $\mathcal{D} = \{(a_i, s_i, v_i)\}_{i=1}^N$ , comprising audio performances  $a_i$ , their symbolic score annotation  $s_i$ , and information regarding the version  $v_i$  in the form of an identifier for the recording or performance. We represent a version condition  $v_i$  as an integer number, where recordings performed by the same ensemble in the same recording environment are assigned the same number (Section IV-C). With this dataset, we train a music synthesizer using different state-of-the-art architectures, namely T5 Transformer [17] or U-Net [11], infused with version conditioning.

Similar to previous works [17], we operate in the spectral domain, using a mel-spectogram representation. Thus,  $a_i$  is represented as a mel-spectrogram of dimensions  $T_i \times F_{mel}$  where  $T_i$  is the number of time frames and  $F_{mel}$  is the number of mel-spectrogram bins. We postulate our method can be adapted to higher-dimensional spectral representations (e.g., STFT), or the waveform domain, though at significantly higher computational costs. To convert mel-spectrograms into audio, we rely on the state-of-the-art Soundstream vocoder [35], which is the same vocoder used by Hawthorne et al. [17]. It was trained with a combination of spectral reconstruction and adversarial losses.

We represent the musical score  $s_i$  as a MIDI-like roll aligned with the audio on the note onset level, which is a matrix of dimensions  $T_i \times (2 \cdot P \cdot (I+1))$ , where  $T_i$  is the number of time frames, and P, I are the numbers of pitches and instruments, respectively. We use additional P classes representing instrument-independent pitches, hence the factor of I + 1. The instrument-independent pitch classes serve two purposes: Learning fundamental frequencies, which are common across instruments, and enabling generation from pitch-only input by applying dropout to the instrument information. Furthermore, we use both onset information and approximate duration for each pitch and instrument, hence the factor of 2.

Finally, to generate long audio streams in a segment-wise fashion, while ensuring seamless transitions between generated segments, we adapt an overlapping technique, borrowed from visual generation (Section V). This technique is complemented by version conditioning, which also implicitly enforces consistency across different generated segments, thus preventing timbre drift.

#### A. Denoising Diffusion Probabilistic Models

We train our neural synthesizer as a Denoising Diffusion Probablistic Model (DDPM) [11]. DDPMs are trained to estimate the inverse of a Gaussian diffusion noising process, parameterized by a noise level schedule  $\beta_t$ , t = 0, ..., T, typically increasing from 0 to 1. By substituting noise level  $\beta_t$  with signal level,  $\alpha_t = 1 - \beta_t$ , the process can be represented as a Markovian chain, in the following recursive form:

$$x_t = \sqrt{\alpha_t} x_{t-1} + \sqrt{1 - \alpha_t} \epsilon \tag{1}$$

where  $t \ge 1$ ,  $x_0$  is a datapoint, i.e. a clean spectrogram, and  $\epsilon \sim \mathcal{N}(0, I)$ . By further substituting  $\bar{\alpha}_t = \prod_{i \le t} \alpha_i$ , one can equivalently represent the process by the non-recursive form:

$$x_t = \sqrt{\bar{\alpha}_t} x_0 + \sqrt{1 - \bar{\alpha}_t} \epsilon.$$
(2)

A DDPM  $\epsilon_{\theta}(x_t, t)$  estimates the reverse process by predicting the normal noise  $\epsilon$ , given the noisy spectrogram  $x_t$  and the diffusion timestep t. In our case of multi-aspect-conditioned music synthesis, it can be trained by minimizing the empirical loss:

$$\mathbb{E}_{(x_0,s,v)\sim\mathcal{D},\,\epsilon\sim\mathcal{N}(0,I),\,t\sim U(\{1,\dots\,T\})}\|\epsilon_\theta(x_t,t,s,v)-\epsilon\|_1 \quad (3)$$

where s, v are the score and version conditions, respectively. In visual models the  $L^2$ -norm is common, however we follow Hawthorne et al. [17] and use the  $L^1$ -norm.

Note that  $\bar{\alpha}_0 = 1$ ,  $\bar{\alpha}_T = 0$ , and  $x_T \sim \mathcal{N}(0, I)$ . Therefore, the model learns to sample from the latent data distribution p by mapping (nearly) isotropic noise to data points:

$$p(x_0|x_T, s, v), \quad x_T \sim \mathcal{N}(0, I), \tag{4}$$

thus modelling the variation in the data. In our approach to multiaspect-conditioned music synthesis, we use DDPMs to capture the variations in musical performances and to account for subtle nuances, since the same musical score can have infinitely many



Fig. 3. Architecture of the T5 Transformer. The difference from Hawthorne et al. [17] is incorporation of version conditioning using FiLM layers. In the spectrogram decoder the version representation is concatenated to the diffusion timestep representation, and both are inserted into FiLM layers. In the score encoder, the version representation is inserted through FiLM layers. Note that the score encoding is independent of the diffusion timestep, similar to Hawthorne et al. [17].

interpretations, even when played by the same musicians under identical acoustic conditions.

# B. Architecture

We experiment with two architectures: A T5 Transformer used for score-conditioned spectrogram synthesis [17], and a U-Net originally used for images [11]. We enhance both models with version conditioning. The T5 architecture, which is our main focus, is depicted in Fig. 3.

The **T5**, borrowed from Hawthorne et al. [17], comprises a transformer decoder, which is the generative backbone that denoises the spectrogram, and a transformer encoder, providing a representation of the score condition as auxiliary input to the decoder, for note and instrument control. The decoder receives the encoded score through cross-attention layers. A central motivation for separating denoising from score conditioning in the architecture is modularity-it theoretically enables separate training of each component, even on different datasets, which is common in text-to-image models [12], [13], [14]. Hawthorne et al. [17] train this model on a synthetic dataset, without version conditioning. We incorporate version conditioning into this model, and train it from scratch on real performances alone. Both the original T5 model of Hawthorne et al. [17] trained on synthetic data, and the T5 trained on our dataset, without version conditioning, serve as baselines for comparison, to evaluate both the effect of version conditioning, and the effect of using real training data (Section VI).

Although most of the evaluation in this work (including the listening tests) is done with the T5, we also experimented with a **U-Net**. We use the exact same architecture as Ho et al. [11], but adapt it to spectrogram synthesis by modifying all operations (convolution, attention, and group normalization) to be 1D rather than 2D, regarding the frequencies as channels. This allows for interactions between distant frequencies, inherent in spectrograms (partial frequencies), and is common practice in



Fig. 4. Multi-aspect conditioning. We condition our model both on the score and on the version. The version is defined by a set of recordings by the same performer, in the same recording environment, e.g., a recording of Toscanini performing Beethoven's 5th symphony from 1952. Our model enables to synthesize new scores with the target version. In the latter example, the model can generate a performance of Brahms' 4th Symphony, with the acoustics and style of the 1952 Toscanini recording of Beethoven's 5th symphony.

spectrogram synthesis (e.g., Wang and Yang [26] use a 1D U-Net without diffusion). Results for the U-Net can be found in the Appendix<sup>2</sup> on our project page. As can be seen there, and from a qualitative observation, the T5 achieves slightly better results, but the U-Net requires significantly less training time.

## IV. MULTI-ASPECT CONDITIONING

Our model is conditioned on multiple aspects: The noise level (as is common in diffusion models), the musical score information, and the version. In the following sections we explain how each condition is incorporated.

#### A. FiLM Layers and Diffusion Time Step Condition

A central mechanism we use for conditioning is Feature-wise Linear Modulation (FiLM) layers [23], which are especially suitable for multi-task cases, where a model learns many similar tasks simultaneously. For example, a diffusion model typically learns to denoise with many different noise levels—each noise level can be regarded as a slightly different task. FiLMs enable the different tasks to share most parameters while maintaining flexibility. The features at each level of the network are slightly modified with an affine transformation, according to the condition. Such transformations are also referred to as modulations (not to be confused with a musical modulation of key).

FiLMs are typically used in diffusion models to handle different noise levels, i.e., diffusion timesteps [17]. In our case of synthesis using version conditioning, we further extend the modulations at each layer to control the desired acoustics depending on the version v (Section IV-C).

We apply FiLMs by predicting an affine transformation for each block of the network using multi-layer perceptrons (MLPs). More explicitly, given a condition c (representing a diffusion timestep t, or an ID number v corresponding to the version, or a combination of both), we learn an MLP embedding  $C = M_{\theta}(c) \in \mathbb{R}^{e}$ . For the *i*-th block of the network with input features  $h_{i,j}$ ,  $0 \leq j < n_i$ , we learn linear layers  $L_i^{\text{scale}}, L_i^{\text{shift}}$ :  $\mathbb{R}^{e} \to \mathbb{R}^{n_i}$  and modulate the input features to obtain new features  $h'_i$ :

$$h'_{i,j} = \text{FiLM}(h_{i,j}|\gamma_{i,j},\beta_{i,j}) = \gamma_{i,j}h_{i,j} + \beta_{i,j}$$
(5)

where  $\gamma_i = 1 + L_i^{\text{scale}}(C)$  and  $\beta_i = L_i^{\text{shift}}(C)$ . See Section IV-C for details on combining the version condition with the diffusion timestep condition.

#### B. Score-Based Condition

A necessary requirement for any synthesizer is control over notes and instrumentation in generated performances. When following a data-driven approach, such control requires training data consisting of musical performances paired with their corresponding reference transcriptions. Any score-related aspect we seek to control should be faithfully represented in the corresponding reference transcription. Therefore, control over notes, exact timing, and instrumentation requires training data to have accurate transcriptions of all these aspects.

Music transcription in general, and especially of multiinstrument performances, is known to be a hard problem due to temporal-spectral overlaps and the lack of training data. Thus, previous DDPM-based works resort to artificial data generated by concatenative synthesis, for which transcriptions are trivially available [17], [30]. Unfortunately, this comes at the expense of realism, as the model learns to imitate the sound of a concatenative synthesizer, rather than real musical performances, and thus does not leverage the full generative power of DDPMs.

Recent work by Maman and Bermano [22] shows potential in transcribing general multi-instrument recordings. They propose a unified framework for automatic music transcription, and audio-score alignment based on neural features of transcription models. We show that this transcription approach provides effective score conditioning, enabling to train a diffusion-based synthesizer on large amounts of uncurated real performances of diverse instrumentation, while maintaining control over notes and instrumentation.

1) Alignment vs. Transcription: The score conditions for the synthesizer should be exact score representations of the corresponding audio. These can be obtained in two ways:

- Alignment of an existing digital score representation with the audio
- Transcription predicted by an automatic transcriber

Both approaches have benefits and limitations in terms of robustness and accuracy: Alignment is generally easier than transcription and is constrained to a well-defined set of note events, including instrument information, eliminating confusion between pitches and instruments. Previous work shows that a weak transcriber can still produce an accurate alignment [22], [36]. On the other hand, automatic prediction can be more robust in case of alignment errors, which often occur [22], [36].

From a practical perspective, transcription is more favorable than alignment, as for most musical genres and performances (e.g. jazz or rock music) accurate scores are not easily available,

<sup>&</sup>lt;sup>2</sup>benadar293.github.io/multi-aspect-conditioning/static/pdfs/Appendix.pdf

or do not exist. A "universal" automatic transcriber (for which recent work [22] shows promising results) provides transcriptions for a wider range of musical performances, facilitating the use of massive training data for the synthesizer, similar to large text-to-image models [12], [13], [14].

In this work, we experiment with both approaches and show that they produce comparable results. Due to the major practical benefits of the transcription approach, we perform qualitative listening tests (Sections VI-D-1-VI-D-2) on a synthesizer trained with transcriptions as conditions. In addition, we quantitatively evaluate and compare both score conditioning approaches, and show comparable performance (see Appendix and samples on project page).

2) Score Errors: We note that both approaches may lead to errors, or deviations between the musical performance and the assigned reference notes. However, a small amount of errors will not necessarily negatively impact the model, as the model is trained to generate performances from the distribution of real performances, that do not contain errors, from possibly noisy scores, thus learning to correct errors. Still, investigating the effect of transcription or alignment accuracy on the model's performance is an important direction for future work.

#### C. Version-Based Condition

In musical performances, many aspects other than the musical score itself bear significant influence on the outcome, including both acoustic- and performance-related aspects. Acousticrelated aspects include specific instrument timbre, e.g., the specific kind of guitar used, the singer identity, the church organ register, or more general acoustics such as the room acoustics, the orchestra size, position of the audience or recording device relative to the orchestra, and many more. Performance- or stylerelated aspects include, for example, dynamics and expression the same musical segment can be played in infinitely many styles with varying intensity or fluctuation, e.g., using vibrato, tremolo, or staccato, legato, etc. All of these aspects are reflected in the specific *version* of the musical piece.

A successful synthesizer should take into account the latter aspects associated with the specific version, in addition to the score. Therefore, we condition our generated performances on the target version, in order to obtain its special characteristics. This *version conditioning* enables acoustic and style control, and reduces the ill-posedness of score-conditioned music synthesis. The different aforementioned version-related factors such as timbre, acoustics, and style are thus represented implicitly by the *version condition*. Using this mechanism, we can generate different types of the same instrument (e.g., guitars of different timbres), or different orchestral sounds. We demonstrate this on our project page, by synthesizing the same scores with different orchestras, harpsichords, church organs, and guitars.

We represent a version condition as an integer number  $v_i$ , where recordings performed by the same ensemble in the same recording environment are assigned the same number. Each condition can represent a single recording of a few minutes in the train set (e.g., Segovia playing Albéniz's Capriccio Catalán on the guitar), or a set of recordings of several hours (e.g., of Beethoven's concertos for piano and orchestra performed by Mitsuko Uchida and The Royal Concertgebouw Orchestra).

1) Version Conditioning With FiLM Layers: We incorporate the version condition using FiLM layers, similar to diffusion timesteps. As mentioned in Section IV-A, FiLM layers are suitable for multi-task cases. In the case of a synthesizer trained on many unrelated performances of diverse instrumentation, each performance with its corresponding acoustics and style should be regarded as a slightly different task.

As mentioned in Section III-B, the T5 architecture we use [17] consists of a spectrogram decoder, denoising the spectrogram, and a score encoder, providing a representation of the notes as auxiliary input to the decoder through cross-attention layers. Note that Hawthorne et al. [17] condition only on the diffusion timestep and not on the version. They use FiLMs in the decoder to incorporate the diffusion timestep into denoising, but not to the score encoder; i.e., the score representation is independent of the noise level.

We incorporate the version condition into the decoder and encoder using FiLMs in the following manner: We learn two MLP representations for the version v, and diffusion timestep t, which we denote by  $M_{\theta}(v)$ ,  $N_{\theta}(t) \in \mathbb{R}^{e}$ . For the decoder FiLMs, we concatenate them, and use the combined condition:

$$C = C(v, t) = \operatorname{concat}(M_{\theta}(v), N_{\theta}(t)) \in \mathbb{R}^{2e}$$
(6)

in (5) (see also Fig. 3). For the score encoder FiLMs we use the version condition alone, without the diffusion timestep, as explained above, leading to the condition:

$$C = C(v) = M_{\theta}(v) \in \mathbb{R}^e.$$
(7)

For the U-Net, we use FiLMs to condition each block on the combination of the diffusion timestep and the version, as in (6).

## V. TEMPORAL COHERENCY & SMOOTH TRANSITIONS

We generate long performances of several minutes by segments of  $\sim$ 5 seconds each, dictated by memory constraints. This raises an issue of coherency and consistency between segments. A naïve approach of block-wise synthesis will allow abrupt changes in volume, timbre, expression, ambience, etc. in transition points, due to different trajectories in the diffusion reverse process. Moreover, even with smooth transitions, timbre drift can occur, i.e., changes in timbre between segments. We propose a simple and effective overlapped generation technique for smooth transition between segments, adapted from computer vision.

## A. Smooth Transitions

For smooth transition between segments, we generate segments in a batch with overlaps, smoothly interpolating between consecutive segments in each step of the sampling process. Denote by  $\hat{x}_0^{\text{left}}$ ,  $\hat{x}_0^{\text{right}}$  the overlapping parts of consecutive segments in a predicted batch  $\hat{x}_0$ , such that  $\hat{x}_0^{\text{left}}$  is the end of a segment and  $\hat{x}_0^{\text{right}}$  is the beginning of the consecutive segment (Fig. 2, bottom right). Denote their shape by  $T_{\text{ol}} \times F_{\text{mel}}$  where  $T_{\text{ol}}$  is the number of overlapping frames. We use an interpolation matrix M of the shape  $T_{ol} \times F_{mel}$ , linearly decreasing from 1 to 0 along the time axis and constant across the frequency axis. We update  $\hat{x}_0^{\text{left}}$  and  $\hat{x}_0^{\text{right}}$  to the value of:

$$M \cdot \hat{x}_0^{\text{left}} + (1 - M) \cdot \hat{x}_0^{\text{right}}.$$
 (8)

This update is done in the predicted sample  $\hat{x}_0$  of each step of the reverse process, derived from the predicted noise  $\hat{\epsilon}$  according to the formula  $\hat{x}_0 = (x_t - \sqrt{1 - \bar{\alpha}_t}\hat{\epsilon})/\sqrt{\bar{\alpha}_t}$ . We use an overlap of  $T_{\rm ol} = 32$  frames, corresponding to 0.64 seconds of audio. Borrowed from motion generation [19], [20], this is an effective and convenient approach, performed solely at the sampling stage, requiring no additional training components, contrary to Hawthorne et al. [17].

## B. Acoustic Consistency

Smooth transitions enabled by the proposed interpolation do not guarantee consistent timbre, as timbre could still drift smoothly between segments, as is demonstrated on our project page. As can be observed in the samples, version conditioning is an effective way to address this issue, as it implicitly creates acoustic consistency and stability between segments.

#### VI. EVALUATION

In this section, we discuss the evaluation of our proposed DDPM-based synthesizer. In Section VI-A we define the desired properties of a high-quality synthesizer and explain the general criteria and methods for evaluation. In Section VI-B we present the datasets used for evaluation. In Section VI-D we present the qualitative listening tests we have performed for evaluating the model. In sections VI-E and VI-F we present quantitative results using score metrics based on the Fréchet Audio Distance (FAD) [24], and transcription metrics.

## A. Evaluation Criteria

A high-quality synthesizer should possess both high generative power on the one hand, and elaborate control on the other hand, while producing high-quality realistic sound. This can be broken down into the following required properties:

*a) Realism & Quality:* The synthesizer should generate realistic and high-quality sound. Ideally, the generated performances should be indistinguishable from real musical performances, i.e., pass the Turing Test [37].

*b) Score Control:* This involves the played notes or pitches, instruments, and timing, including note onset and offset. Notice that note onset and offset timing are also related to *performance control*, as discussed below.

*c) Acoustic Control:* This involves the specific instrument timbre (e.g., which type of violin), room acoustics, location of listener or microphone relative to the orchestra, orchestra size, and so on.

d) **Performance Control:** This encompasses aspects such as note intensity or strength (also referred to in the literature as the *velocity* in which the instrument was struck), degree of vibrato

or tremolo, and more. Also, expressive interpretation of note timing can be regarded as a part of performance aspects.

e) Generating Unspecified Aspects: While control over the aforementioned aspects (score, acoustics, performance) is highly desirable, it is also required that the model can realistically generate aspects or confounding factors unspecified by the user, facilitating the generation process. This is required since many aspects of control, especially performance controls, such as vibrato and velocity, can be extremely laborious to define, and require highly skilled musicians. This also enables further flexibility, allowing to generate multi-instrument performances from partial conditions such as pitch-only input without specifying instrumentation, or performances with drums from a musical score without drums, as is demonstrated on our project page.

In this work we focus on *realism and quality, score control*, and *acoustic control*. We rely on the model to *implicitly generate performance aspects* such as vibrato and intensity. Examples can be found on our project page demonstrating the ability of the model to generate such aspects, e.g., a violin playing with vibrato. We argue that explicit performance control in a data-driven approach requires the transcriptions to contain this information, and leave this for future work.

To evaluate *realism and quality*, in Section VI-D-1 we conduct a MUSHRA listening test [38] comparing performances of the same musical excerpts generated on various synthesizers, both concatenative and diffusion-based, with and without version conditioning, as well as real versions, both original and vocoded.

Furthermore, to evaluate *acoustic control*, in Section VI-D-2 we conduct an additional listening test to measure the effect of version conditioning in achieving the target acoustics, to which we refer to as *version similarity*.

In Section VI-E we complement the qualitative listening tests with quantitative perceptual metrics based on the Fréchet Audio Distance (FAD). We introduce two ways of measuring the FAD: *All-FAD* to measure realism and quality, and *Group-FAD* to measure version similarity. Finally, in Section VI-F, to evaluate *score control*, we apply quantitative transcription metrics to evaluate the faithfulness of our generated performances to the target notes and instruments.

As explained in Section IV-B-1, we present here results from a model trained with automatic transcriptions as score conditions due to its higher practicality. We provide quantitative and qualitative results for the model trained with aligned MIDI scores in the Appendix<sup>2</sup> and on our project page.

## B. Datasets

1) Train Set: We train our model on 197 performances of Western classical music, including symphonies, chamber music, and solo pieces, comprising 19 instruments, and totaling 58:25:47 hours. The total duration and the number of versions for each ensemble are shown in Table II. The data consists of performances from YouTube<sup>3</sup> and Musopen,<sup>4</sup> with corresponding MIDI transcriptions from KunstDerFuge,<sup>5</sup> aligned as proposed

<sup>&</sup>lt;sup>3</sup>https://www.youtube.com

<sup>&</sup>lt;sup>4</sup>https://musopen.org

<sup>&</sup>lt;sup>5</sup>https://www.kunstderfuge.com

TABLE II ENSEMBLE DISTRIBUTION OF THE TRAIN SET  $\mathcal{D}_{train}$ , Including the Total Duration for Each Ensemble Given in H:mm:ss, and the Number of Versions Existing for Each Ensemble

Ensemble	Duration	#Versions
Flute & Harpsichord	0:49:19	2
Orchestra	14:31:13	39
Orchestra & Choir	1:49:54	1
Orchestra & Piano	5:12:50	7
Solo Cello	2:16:16	2
Solo Flute	0:09:16	1
Solo Guitar	4:40:14	85
Solo Harpsichord	7:27:40	12
Solo Organ	2:10:49	25
Solo Piano	5:52:41	17
Solo Violin	1:56:49	1
Violin & Harpsichord	3:18:48	2
Violin & Piano	3:59:17	1
Violin, Cello, & Piano	3:37:23	1
Wind Quintet	0:33:10	1
All	58:25:47	197

by Maman and Bermano [22]. Following the same work, we augment the data by pitch-shifting up to  $\pm 2$  semitones (larger ranges of shifts did not improve performance). We label the data with version condition IDs by assigning numerical indices to the different performances, where typically the same index is given to an entire set of recordings (e.g., a CD box with Beethoven's Piano Trios recorded by the same ensemble in the same studio). For church organ performances, where different organ registers are used in different audio tracks, resulting in significant timbre differences, we assign an individual version ID for each track. For the distribution of version lengths for each ensemble we refer to the Appendix<sup>2</sup> available on our project page. It can be seen that version lengths vary between a few minutes (e.g., guitar or organ) and a few hours (e.g., violin & piano duo). As we show in the following, version conditioning enables version control despite possible imbalance in version durations. Samples on the project page demonstrate the ability to successfully condition the model on recordings of a few minutes-the guitar and organ versions, as well as the orchestral version "Czech Symphony Orchestra playing Beethoven's Coriolan Overture" on the project page each contain less than ten minutes. For further information on the data collection process, including the distribution of composers, see Appendix. We denote this dataset  $\mathcal{D}_{train}$ .

2) Listening Tests Evaluation Set: For the listening tests, we use 12 MIDI performances of Western classical pieces, none of which appear in the train set, but containing the same instruments as in  $\mathcal{D}_{train}$ . The pieces include orchestral music, wind quintet, church organ, and harpsichord pieces. We use the first 3:00 minutes of each MIDI performance, which yields a total duration of 36:00 minutes. Short excerpts used for the listening tests were drawn randomly from these performances. As references for comparison, for each MIDI performance we use a corresponding real musical performance of the same score, of the same instrumentation, but of a version that does not appear in  $\mathcal{D}_{train}$ . See Appendix for the list of pieces in this dataset. We denote this dataset  $\mathcal{D}_{listen}$ .

3) Large Quantitative Evaluation Set: We quantitatively evaluate our models with 58 MIDI performances of Western classical pieces of a total duration of 5:09:30 hours, none of which appear in the train set, but containing the same instruments as in  $\mathcal{D}_{\text{train}}$ . For each test MIDI, we randomly sample three version conditions for synthesis. For example, the test MIDI can be a performance of Mozart's 40th symphony, and the version condition can be the performance of the Berlin Philharmonic Orchestra playing Brahms' Haydn Variations. This is the same evaluation set used by Maman et al. [21]. See Appendix for the ensemble distribution of this dataset. We denote this dataset  $\mathcal{D}_{quant}$ .

# C. Training & Sampling

We use a T5 backbone similar to the "base" model used by Hawthorne et al. [17], extending it with FiLM layers for conditioning on the version. The model has 12 layers in the encoder and decoder, with 12 attention heads of dimension 64 each, resulting in 768 feature dimensions. For the attention layers we use 2048 feedforward dimensions. We trained the model with batch size 312 using five-second segments, for 300K steps. This took approximately 350 hours on three Nvidia A100 GPUs. We used an Adam [39] optimizer with learning rate 1e-4, reducing the learning rate to 1e-5 for the last 30K steps. For details on the U-Net see Appendix.

We train the model with 1000 diffusion timesteps, using a cosine noise schedule [40]. We perform DDIM [41] sampling of five-second segments using 250 steps. To control the condition strength, we use classifier-free guidance (CFG) [42] with weights 1.25 for both the score and the version. For further information on CFG with multiple conditions, see Appendix.

## D. Evaluation Based on Listening Tests

We performed two distinct listening tests, to evaluate both realism (Section VI-D-1) and similarity to the target version (Section VI-D-2). Both listening tests are publicly available on our project's listening test page<sup>6</sup>.

1) Realism Listening Test: In this listening test, we aim to evaluate to what degree our generated performances resemble real musical performances, as opposed to synthesized ones. Ideally, generated performances should be indistinguishable from authentic recordings. For this, we follow an adaptation of the MUSHRA (Multiple Hidden Stimuli with Hidden Reference and Anchor) protocol [38] to evaluate realism of generated performances.

MUSHRA is a protocol for assessing audio quality, originally created for evaluating audio compression. Typically, the same excerpt is provided multiple times, restored with different codecs or algorithms. Each comparison of excerpts contains both a reference sample and one or more anchor samples. The reference represents the ideal quality, i.e., no compression is applied. The anchors represent baselines, against which different methods can be compared, and are used to calibrate the rating scale. The listener is provided with the reference sample and the set of samples for evaluation, also including a hidden version of the reference sample. The listener is asked to rate all samples on a

<sup>6</sup>https://benadar293.github.io/listening-tests

scale from 0 to 100, and is required to rate the hidden reference with a score 90-100 (depending on the standard that is used). The MUSHRA protocol is considered robust, enabling statistically meaningful results with relatively few participants.

We apply the same methodology to evaluate the *realism* of our generated performances. Specifically, we want to find out if and to what extent the generated performance sounds like a real musical performance, as opposed to one performed on a synthesizer. The listeners are presented with an audio excerpt from a real musical performance as a reference, which represents ideal realism, and multiple synthesized versions of the same score content appearing in the excerpt, using different models or methods. As lower anchors we use synthesized versions obtained by concatenative synthesis, using two different soundfonts: The standard Windows Media Player GM soundfont, and the Fluid Release 3 General-MIDI soundfont, which is of higher quality. Concatenative synthesis serves as a baseline for achievable realism. We also include the vocoded version of the reference, in order to isolate the influence on realism of the vocoder from the spectrogram synthesizer. The vocoded version was obtained by calculating the mel-spectrogram of the reference sample and restoring the audio with the same vocoder we use for synthesis. Note that the vocoder quality is an *upper bound* on the quality achievable by our method. We compare excerpts of the same score content generated by the following neural synthesizers:

- 1) The original T5 model of Hawthorne et al. [17], which was trained mainly on data produced by a concatenative synthesizer, without version conditioning.
- 2) Our T5 model trained on our dataset, without version conditioning.
- Our T5 model trained on our dataset, with version conditioning.

Note that the latter three configurations are all based on the same architecture. The differences are the type of data, real or synthetic, and whether or not version conditioning is used.

In total, seven versions were compared in each question: Two from concatenative synthesizers, three from neural synthesizers, the vocoded version, and the hidden reference. The seven versions were presented in random order. We asked the listeners to rate the realism of each excerpt from 0 (unrealistic) to 100 (realistic) and informed the listener that there is a hidden reference that must be ranked close to 100. The listener had to rate the reference with a score above 90 more than 85% of the time, otherwise the questionnaire was discarded.

The test comprised 32 participants, five of whom were discarded, leaving 27 participants. The questionnaire included ten questions. In each question, a twelve-second random excerpt was drawn from a MIDI sampled from  $\mathcal{D}_{listen}$ , and sonified in the seven aforementioned variations (concatenative synthesizers, diffusion model, vocoder, etc.). The reference excerpt was obtained from a real performance of the same piece as the MIDI, through automatic alignment, which was manually verified to ensure it contained the same content. The test was conducted using the webMUSHRA implementation by the webMUSHRA implementation [43].

Results can be seen in Fig. 5 and Table III. It is evident that our model trained on real performances produces more realistic



Fig. 5. MUSHRA realism listening test results in box plot form. For the exact Mean Opinion Scores (MOS) appearing in this figure, see Table III. 'GM' and 'Fluid' are soundfonts used for concatenative synthesis. 'Hawth.' is the model of Hawthorne et al. [17]. 'Uncond.' is our model without version conditioning (but with score conditioning). 'Cond.' is our model with both version conditioning and score conditioning. 'Vocoded' is the reference sample, after applying the vocoder to its mel-spectrogram, and is an upper bound on the achievable realism. It can be seen that both our models ('Cond.' and 'Uncond.') produce significantly more realistic performances than all compared models and methods.

TABLE III REALISM MUSHRA LISTENING TEST MEAN OPINION SCORES (MOS)

Model / Method	Abbreviation	Realism MOS ↑
Windows GM Soundfont	GM	24.0
Fluid R3 GM Soundfont	Fluid	44.5
Hawthorne et al. [17]	Hawth.	30.7
Ours w/o version condition	Uncond.	48.4
Ours w/ version condition	Cond.	52.1
Vocoded	Vocoded	86.2
Reference	Real	99.0

These are the results appearing in fig. 5 in box plot form. We list the abbreviations appearing in fig. 5.

results than all the baselines-both concatenative synthesizers (GM, Fluid), and the model of Hawthorne et al. [17], by a significant margin. This holds whether or not using version conditioning. For example, the basic concatenative synthesizer (GM) obtains 24.0 MOS in realism, where our model with version conditioning obtains 52.1 MOS. Our model also surpasses the more advanced concatenative synthesizer (Fluid), improving MOS from 44.5 to 52.1. Noteably, although we use the same architecture as Hawthorne et al. [17], the difference in realism is significant-over 20 MOS difference, from 30.7 to 52.1, due to the difference in training data. Also note that generating with or without version conditioning yields comparable realism scores, where version conditioning slightly improves MOS from 48.4 to 52.1. We note in this context that version conditioning mainly influences version similarity, discussed in Section VI-D-2. Lastly, note that the vocoder quality has a non-negligible effect on the final result, as merely restoring the audio from the mel-spectrogram with the vocoder reduces the score from 99.0 to 86.2.

We conclude from these results that using real musical performances in a diffusion-based approach for music synthesis significantly improves realism compared to data based on traditional (concatenative) synthesis.



Fig. 6. Similarity listening test results. We synthesize the same score excerpt using our model, with three different version conditions, which are of performances with the same instrumentation. One of the three serves as a reference version. We ask the user to rate the similarity of each of the synthesized excerpts to a random excerpt of different score content, sampled from the reference version. Left ('Other'): Similarity to the reference version of excerpts synthesized with version conditions other than the reference version. Right ('Reference'): Similarity to the reference version condition.

2) Version Similarity Listening Test: In this listening test, we are interested in evaluating if and to what extent our generated performances perceptually resemble the reference version used for conditioning, in terms of acoustics, timbre, performance, style, etc. For example, if generating an organ piece with a version condition of a specific organ, we want to evaluate to what extent the organ in the generated performance sounds like the organ given in the reference version.

In this listening test, for each comparison we randomly sample a reference audio excerpt  $a_{ref}$  with version  $v_{ref}$  and score  $s_{ref}$ from  $\mathcal{D}_{\text{train}}$ . We sonify a short score excerpt s (in MIDI form, of different content than the reference  $s_{ref}$ ) three times, once with the same version condition  $v_{ref}$  corresponding to the reference, and with two additional randomly sampled version conditions  $v_1, v_2$ , of the same instrumentation as  $v_{ref}$ . We denote the outputs of the model by  $a_{ref}^s$ ,  $a_1^s$ ,  $a_2^s$ . For example, following Fig. 4, our reference  $(a_{ref}, v_{ref}, s_{ref})$  is Toscanini's performance of Beethoven's 5th Symphony. As a score condition s we use an excerpt from Brahms' 4th Symphony, which we sonify with the reference version condition  $v_{ref}$  being Toscanini's performance of Beethoven's 5th Symphony, and additional version conditions  $v_1, v_2$  being Karajan's performance of Brahms' 2nd Symphony, and Barenboim's performance of Mozart's 41st Symphony. For the corresponding synthesized excerpts  $a_{ref}^s, a_1^s, a_2^s$ , we ask the user to rate the similarity of each to the reference excerpt  $a_{ref}$ , on a scale from 0 to 100. The similarity test comprised 26 participants and included ten questions. For each question, we sampled a random twelve-second excerpt from a MIDI sampled from  $\mathcal{D}_{listen}$ , and sonified it with the three aforementioned version conditions, comparing them to the reference excerpt.

Results appear in Fig. 6. In the left part, it can be seen that version similarity scores improve significantly by conditioning on the reference version, compared to conditioning on other versions, even though they are of the same instrumentation. The improvement is over 20 in the MOS, from 51.7 to 72.0 on average.

These results show that the notion of version similarity, reflected in perceptual and acoustic similarity, is indeed meaningful, and such similarity can be achieved by using version conditioning. The results clearly indicate that version conditioning is an effective means to obtain version-specific characteristics, e.g., specific timbre or acoustics that appear in a target version.

# E. Quantitative Evaluation Using Fréchet Audio Distance

We complement the listening tests with a quantitative evaluation using the *Fréchet Audio Distance (FAD)* [24]—a perceptual score with origins in computer vision [44]. We use this metric in different ways to quantitatively evaluate realism and quality (Section VI-E-1), and version similarity, i.e., the resemblance of our generated performances to the conditioning version (Section VI-E-2).

FAD relies on large models based on deep neural networks, such as TRILL [45], trained on large real-world datasets to predict embedding vectors from snippets of input audio. The assumption is that perceptually similar audio snippets yield closely spaced embedding vectors. To compute FAD between two audio datasets  $\mathcal{D}_1$ ,  $\mathcal{D}_2$  (e.g., a set  $\mathcal{D}_1$  of synthesized audio conditioned on a specific version, and a set  $\mathcal{D}_2$  of real recordings of the same version), the mean vectors  $\mu_1$ ,  $\mu_2$  and the covariance matrices  $\Sigma_1$ ,  $\Sigma_2$  are computed over all embedding vectors generated from  $\mathcal{D}_1$  and  $\mathcal{D}_2$ , respectively. The FAD is then defined as:

$$FAD(\mathcal{D}_1, \mathcal{D}_2) = |\mu_1 - \mu_2|^2 + tr\left(\Sigma_1 + \Sigma_2 - 2(\Sigma_1 \Sigma_2)^{1/2}\right).$$
(9)

Intuitively, if  $D_1$ ,  $D_2$  are perceptually similar, the distributions of the model's responses over the two datasets should be similar, resulting in a small distance. Kilgour et al. [24] show that FAD correlates with human perception and that increasing distortions increase the FAD. The results we present in the following sections further confirm this, as we show that qualitative listening test results are consistent with FAD scores.

We use two models as backbones for FAD, also used by Hawthorne et al. [17] for evaluation: TRILL [45] (5.9 embeddings/sec.), and VGGish [46] (1 embedding/sec.). We measure FAD in two ways, differing in the choice of the compared datasets: *All-FAD* for realism and quality (Section VI-E-1), and *Group-FAD* for version similarity (Section VI-E-2).

We report quantitative metrics for the generated performances used in the listening tests, and show they are consistent with the qualitative listening tests' results. For statistical stability, the metrics were measured on the entire  $\mathcal{D}_{listen}$  evaluation set from which excerpts in the listening tests were sampled. In addition, for further statistical stability, we measure quantitative metrics on the large scale 5-hour evaluation set  $\mathcal{D}_{quant}$  described in Section VI-B, also used by Maman et al. [21], and compare generation with and without version conditioning.

1) All-FAD—Realism & Quality: To assess realism, quality, and fidelity, we use FAD comparing the entire synthesized evaluation set to the entire train set. In the terms of (9), we define  $\mathcal{D}_1 = f_{\theta}(\mathcal{D}_{\text{listen}}^{\text{midi}})$ , where  $f_{\theta}(\mathcal{D}_{\text{listen}}^{\text{midi}})$  denotes our DDPM-based synthesizer  $f_{\theta}$  applied to the set of MIDIs in  $\mathcal{D}_{\text{listen}}$ , and

	All-FAD↓		
Method	VGGish	TRILL	
$\mathcal{D}_{ ext{listen}}$	Evaluation	Set	
GM	10.07	0.47	
Fluid	12.35	0.67	
Hawth. [17]	7.7	0.33	
Uncond.	4.03	0.18	
Cond.	4.81	0.20	
Vocoded	5.68	0.17	
Real	2.85	0.15	
$\mathcal{D}_{\mathrm{quant}}$	Evaluation	Set	
Uncond.	3.05	0.12	
Cond.	3.58	0.11	

 $\mathcal{D}_2 = \mathcal{D}_{train}^{audio}$ , where  $\mathcal{D}_{train}^{audio}$  denotes the audio recordings in the train set  $\mathcal{D}_{train}$ . Note that  $\mathcal{D}_{listen}^{midi}$  can also be replaced by  $\mathcal{D}_{quant}$ . Also note that  $\mathcal{D}_{train}$ ,  $\mathcal{D}_{listen}$  contain both audio and scores, while  $\mathcal{D}_{quant}$  contains only scores.

This captures the general similarity of the synthesized performances to real performances (or more specifically, to the entire set of real performances used for training), rather than resemblance to a specific version. We refer to this metric as *All-FAD*. While this metric is important for measuring realism and quality, note that the main quantitative metrics for evaluating version conditioning are the *Group-FAD* and *classification accuracy* metrics discussed later.

Results for the All-FAD metric appear in Table IV (lower is better). It can be seen that using real training data significantly improves All-FAD w.r.t. all baselines, whether or not using version conditioning (Cond., Uncond.). For example, the All-FAD based on VGGish is over 10.0 for concatenative synthesizers and 7.7 for the model of Hawthorne et al. [17]. It improves to 4.03 (Uncond.) or 4.81 (Cond.) when using our model. This is consistent with the realism listening test results (Fig. 5, Table III). Note that the reference data  $\mathcal{D}_{\text{listen}}^{\text{audio}}$  (Real), i.e., the audio from  $\mathcal{D}_{\text{listen}}$  and  $\mathcal{D}_2 = \mathcal{D}_{\text{train}}^{\text{audio}}$  in (9)), despite being from versions not appearing in the train set. This implies that the All-FAD metric indeed reflects realism and quality, beyond mere similarity to the train set.

Results on the large evaluation set  $\mathcal{D}_{quant}$  show that version conditioning does not significantly affect All-FAD, implying that general quality is maintained. We also observe that version conditioning might slightly increase the All-FAD. This can be interpreted as follows: Version conditioning shifts the distribution of the generated performances from the general distribution of the train set  $\mathcal{D}_{train}$ , towards the distribution of specific versions, as explained in Section VI-E-2.

2) Group-Fad: To measure version similarity, i.e., how well generated performances resemble the target conditioning version, we introduce the Group-FAD metric. To motivate its use, we show in Fig. 7 a t-Distributed Stochastic Neighbor Embedding (t-SNE) visualization [47] from the train set's TRILL embedding distribution. Each point represents the mean embedding of an audio track (e.g., a movement in a symphony), and each color represents a recording, comprising multiple such tracks, corresponding to a specific version. It can be seen that tracks



Fig. 7. T-SNE Visualization of the TRILL embedding space. Points represent audio tracks, and colors represent complete recordings of specific version IDs, comprising multiple such tracks.

 TABLE V

 Results of the Group-FAD Metric, for Listening Test Performances

	Group-FAD↓			
Model	VGGish	TRILL		
$\mathcal{D}_{ ext{listen}}$	$\mathcal{D}_{\text{listen}}$ Evaluation Set			
GM	15.15	0.90		
Fluid	15.35	1.10		
Hawth. [17]	13.55	0.77		
Uncond.	7.61	0.50		
Cond.	5.15	0.30		
Vocoded	8.69	0.45		
Real	6.03	0.44		
$\mathcal{D}_{\text{quant}}$ Evaluation Set				
Uncond.	7.46	0.55		
Cond.	5.68	0.36		

TABLE VI Results of Version Classification Based on Group-FAD, for Listening Test Performances

	Classification Accuracy				
Model	Top-1	Top-3	Top-5		
$\mathcal{D}_{l}$	$\mathcal{D}_{\text{listen}}$ Evaluation Set				
GM	9.1%	9.1%	9.1%		
Fluid	0.0%	0.0%	0.0%		
Hawth. [17]	18.2%	18.2%	18.2%		
Uncond.	18.2%	27.3%	45.5%		
Cond.	81.8%	90.9%	100.0%		
Vocoded	27.3%	36.4%	45.5%		
Real	45.5%	45.5%	63.6%		
$\mathcal{D}_{ ext{quant}}$ Evaluation Set					
Uncond.	16.8%	31.6%	41.9%		
Cond.	66.5%	83.9%	88.4%		

of the same version form clusters. Following this insight, we define the *Group-FAD* metric: To measure how well our versionconditioned synthesized performances resemble the target version in acoustics, timbre, etc., we compute FAD comparing each score *s* synthesized with a version condition *v*, to the subset of the training set recordings corresponding to *v*, which we denote  $\mathcal{D}_v^{\text{audio}}$ . In the terms of (9), we define  $\mathcal{D}_1 = f_\theta(s, v)$ , where  $f_\theta$ denotes our DDPM-based synthesizer, and  $\mathcal{D}_2 = \mathcal{D}_v^{\text{audio}}$ .

Group-FAD results appear in Table V. It can be seen that version conditioning dramatically and consistently improves Group-FAD, both compared to the baselines, and compared to our model without version conditioning. Note for example, that Group-FAD based on VGGish improves from 7.61 when using our model without version conditioning, to 5.15 when using our model with version conditioning. Notably, when using version conditioning, Group-FAD is better than that of the real samples  $\mathcal{D}_{\text{listen}}^{\text{audio}}$  playing the same score but of another version (Real). For example, the TRILL-based Group-FAD for the real samples is 0.44, and improves to 0.3 for our version-conditioned samples (as explained in Section VI-D-1, the reference samples  $\mathcal{D}_{\text{listen}}^{\text{audio}}$ are from real performances of the test MIDIs, of versions that do not appear in the train set). Results are consistent between VGGish and TRILL, and are consistent with the version similarity listening test (Section VI-D-2), indicating perceptual similarity to the target version obtained by version conditioning.

3) Version Classification: We further use the Group-FAD metric to classify the version of generated performances according to the Group-FAD nearest neighbor, over all training versions. In an initial experiment on version classification of *real recordings*, TRILL was more accurate than VGGish, therefore we use TRILL for version classification of generated performances.

Results are shown in Table VI. Note, for example, that version conditioning (Cond.) improves version classification accuracy by 35-60%. The improvement is dramatic both compared to performances generated w/o version conditioning (Uncond.), and also compared to real performances (Real)—from 18.2% and 45.5% respectively to 81.8%.

## F. Score Control Evaluation Using Transcription Metrics

In this experiment, we quantitatively evaluate whether our DDPM-based model accurately renders the score as specified by the input score (given as MIDI representation). To this end, we use transcription metrics to determine if the generated performances contain the correct notes at the correct times, played by the correct instruments. We assess the transcription accuracy of the synthesized performances using an automatic transcriber [22], [25], trained on the same data as the synthesizer. We compare the note events in the input score to those in the transcription of the synthesized performance, and measure the F1 score for the following:

- Note: Accuracy of pitch and onset within 50 ms.
- Note-with-Instrument: Accuracy of pitch, onset within 50 ms, and correct instrument.
- Frame: Accuracy of note duration.

Results appear in Table VII. It can be seen that all methods produce transcription metrics that are on a rather comparable scale, except for the note-with-instrument metric. Our model (Cond., Uncond.) reaches a note-level accuracy of  $\sim$ 65–67%, which is of reasonable magnitude when considering the complexity of highly polyphonic orchestral music. Note that the transcription metrics are influenced not only by the synthesizer's quality but also by the transcriber. The transcriber was trained on the same data as the synthesizer, and therefore might not perform as well on data generated by other synthesizers. Results on the large evaluation set show comparable transcription accuracy whether or not using version conditioning, similar to the All-FAD metric.

TABLE VII TRANSCRIPTION RESULTS, FOR LISTENING TEST PERFORMANCES ('NOTE+IN.' IS THE NOTE-WITH-INSTRUMENT METRIC)

	Transcription F1% ↑			
	Note	Note+In.	Frame	
$\mathcal{D}_{\text{listen}}$ Evaluation Set				
GM	60.0	27.0	59.2	
Fluid	56.2	39.5	59.2	
Hawth. [17]	54.3	16.4	58.8	
Uncond.	65.7	45.8	59.3	
Cond.	65.0	48.4	61.5	
$\mathcal{D}_{ ext{quant}}$ Evaluation Set				
Uncond.	66.9	50.7	64.8	
Cond.	64.7	46.2	63.9	

In the qualitative listening tests (Sections VI-D-1, VI-D-2), we focus on realism and version similarity, and not on score control. Although we provide quantitative results for score control, further listening tests for this aspect are an important direction for future work.

#### VII. CONCLUSIONS AND FUTURE WORK

We presented a framework for training neural diffusion-based synthesizers on real uncurated musical performances with diverse acoustics and instrumentation, including orchestral symphonies. Our proposed approach is based on diffusion models with multi-aspect conditioning, on both score and version. Through qualitative listening tests and quantitative evaluations, we have demonstrated that our approach produces performances with improved realism, and provides novel acoustic control, enabling to generate performances with version-specific characteristics, such as timbre and room acoustics. We strongly believe our approach is a significant step towards hyper-realistic and controlled music synthesis. Aside from extension to other genres, such as jazz, ethnic, and pop music, there are several important directions for future work, a few of which we outline:

#### 1) Instrumentation Generation

Initial results provided on our project page demonstrate the ability to generate instrumentation from pitch-only input, implicitly controlling instrumentation through versionconditioning. Further investigation of this matter is an important direction for future work.

#### 2) Unseen Versions

Our model can generate performances of unseen scores, with versions from the train set. Adapting a model to new unseen versions through inference from an example excerpt, or test-time adaptation, is highly desirable. Initial results on our project page indicate that conditioning on the version using TRILL embeddings is an effective means for handling unseen versions, at least for known instruments. We plan to further investigate this in future work.

## 3) Version Embedding Space & Interpolation

We provide on our project page samples for performances generated by interpolating between different versions using different techniques—both interpolating between predictions, and interpolating between the learned version embeddings. Interpolating between versions relates to learning a structured and meaningful version embedding space, which is highly desired, and also to handling unseen versions, mentioned above. A thorough investigation of this matter is left for future work.

# 4) Human Singing Voice & Lyrics Conditioning

While in this work we focus on instrumental music, we believe a unified diffusion-based framework for music and human speech is possible. In particular, we believe our approach could be applied to human singing voice synthesis, by applying similar additional conditioning techniques on lyrics or phonemes, and singer or speaker ID.

# 5) Elaborate Performance Control

In this work, we rely on the model to generate performance aspects such as note intensity and vibrato. Adding conditions for such performance aspects will enhance control, but will require enhancement of the transcriptions to comprise these aspects, which is a field of ongoing research.

## 6) Other Spectral Representations

In this work, we focus on mel-spectrogram synthesis and use a vocoder to convert the spectrogram to audio. We believe our approach can be applied to other spectral representations such as the magnitude-STFT or even the complex STFT, with higher computational costs.

#### REFERENCES

- D. Schwarz, "Current research in concatenative sound synthesis," in *Proc. Int. Comput. Music Conf.*, Barcelona, Spain, Sep. 2005.
- [2] D. Schwarz, "Concatenative sound synthesis: The early years," J. New Music Reaserch, vol. 35, no. 1, pp. 3–22, Mar. 2006.
- [3] B. L. Sturm, "Adaptive concatenative sound synthesis and its application to micromontage composition," *Comput. Music J.*, vol. 30, no. 4, pp. 46–66, 2006.
- [4] E. Maestre, R. Ramírez, S. Kersten, and X. Serra, "Expressive concatenative synthesis by reusing samples from real performance recordings," *Comput. Music J.*, vol. 33, no. 4, pp. 23–42, 2009.
- [5] K. Karplus and A. Strong, "Digital synthesis of plucked-string and drum timbres," *Comput. Music J.*, vol. 7, no. 2, pp. 43–55, 1983.
- [6] J. O. Smith, "Physical modeling using digital waveguides," *Comput. Music J.*, vol. 16, no. 4, pp. 74–91, 1992.
- [7] I. J. Goodfellow et al., "Generative adversarial networks," *Commun. ACM*, vol. 63, pp. 139–144, 2020.
- [8] D. P. Kingma and M. Welling, "Auto-encoding variational bayes," in *Proc.* 2nd Int. Conf. Learn. Representations, Y. Bengio and Y. LeCun, Eds. Banff, AB, Canada, Apr. 2014. [Online]. Available: http://arxiv.org/abs/1312. 6114
- [9] T. Karras, S. Laine, and T. Aila, "A style-based generator architecture for generative adversarial networks," in *Proc. IEEE Conf. Comput. Vis. Pattern Recognit.*, 2019, pp. 4401–4410.
- [10] T. Karras, S. Laine, M. Aittala, J. Hellsten, J. Lehtinen, and T. Aila, "Analyzing and improving the image quality of stylegan," in *Proc. 2020 IEEE/CVF Conf. Comput. Vis. Pattern Recognit.*, Seattle, WA, USA, 2020, pp. 8107–8116.
- [11] J. Ho, A. Jain, and P. Abbeel, "Denoising diffusion probabilistic models," in Proc. Adv. Neural Inf. Process. Syst., 2020, pp. 6840–6851.
- [12] A. Radford et al., "Learning transferable visual models from natural language supervision," in *Proc. Int. Conf. Mach. Learn.*, 2021, pp. 8748–8763.
- [13] R. Rombach, A. Blattmann, D. Lorenz, P. Esser, and B. Ommer, "High-resolution image synthesis with latent diffusion models," in *Proc. IEEE/CVF Conf. Comput. Vis. Pattern Recognit.*, New Orleans, USA, 2022, pp. 10674–10685.

- [14] C. Saharia et al., "Photorealistic text-to-image diffusion models with deep language understanding," in *Proc. Adv. Neural Inf. Process. Syst. 35: Annu. Conf. Neural Inf. Process. Syst.*, New Orleans, LA, USA, 2022, pp. 36479–36494.
- [15] M. Jeong, H. Kim, S. J. Cheon, B. J. Choi, and N. S. Kim, "Diff-TTS: A denoising diffusion model for text-to-speech," in *Proc. 22nd Annu. Conf. Int. Speech Commun. Assoc.*, Brno, Czechia, 2021, pp. 3605–3609.
- [16] V. Popov, I. Vovk, V. Gogoryan, T. Sadekova, and M. A. Kudinov, "Grad-TTS: A diffusion probabilistic model for text-to-speech," in *Proc. Int. Conf. Mach. Learn.*, 2021, vol. 139, pp. 8599–8608.
- [17] C. Hawthorne et al., "Multi-instrument music synthesis with spectrogram diffusion," in *Proc. Int. Soc. Music Inf. Retrieval Conf.*, 2022, pp. 598–607.
- [18] Q. Huang et al., "Noise2Music: Text-conditioned music generation with diffusion models," 2023, *arXiv:2302.03917*.
- [19] J. Tseng, R. Castellon, and C. K. Liu, "Edge: Editable dance generation from music," in *Proc. IEEE/CVF Conf. Comput. Vis. Pattern Recognit.*, 2023, pp. 448–458.
- [20] Y. Shafir, G. Tevet, R. Kapon, and A. H. Bermano, "Human motion diffusion as a generative prior," in *Proc. 12th Int. Conf. Learn. Representations*, 2024. [Online]. Available: https://openreview.net/forum?id=dTpbEdN9kr
- [21] B. Maman, J. Zeitler, M. Müller, and A. H. Bermano, "Performance conditioning for diffusion-based multi-instrument music synthesis," in *Proc. IEEE Int. Conf. Acoust., Speech, Signal Process.*, Seoul, South Korea, 2024, pp. 5045–5049.
- [22] B. Maman and A. H. Bermano, "Unaligned supervision for automatic music transcription in the wild," in *Proc. Int. Conf. Mach. Learn.*, Baltimore, Maryland, USA, 2022, pp. 14918–14934.
- [23] E. Perez, F. Strub, H. de Vries, V. Dumoulin, and A. C. Courville, "Film: Visual reasoning with a general conditioning layer," in *Proc. AAAI Conf. Artif. Intell.*, 2018, pp. 3942–3951.
- [24] K. Kilgour, M. Zuluaga, D. Roblek, and M. Sharifi, "Fréchet audio distance: A reference-free metric for evaluating music enhancement algorithms," in *Proc. Annu. Conf. Int. Speech Commun. Assoc.*, Graz, Austria, 2019, pp. 2350–2354.
- [25] C. Hawthorne et al., "Enabling factorized piano music modeling and generation with the MAESTRO dataset," in *Proc. Int. Conf. Learn. Representations*, New Orleans, Louisiana, USA, 2019.
- [26] B. Wang and Y. Yang, "Performancenet: Score-to-audio music generation with multi-band convolutional residual network," in *Proc. Conf. Artif. Intell.*, Honolulu, Hawaii, 2019, pp. 1174–1181.
- [27] J. W. Kim, R. M. Bittner, A. Kumar, and J. P. Bello, "Neural music synthesis for flexible timbre control," in *Proc. IEEE Int. Conf. Acoust., Speech, Signal Process.*, Brighton, U.K., 2019, pp. 176–180.
- [28] H. Dong, C. Zhou, T. Berg-Kirkpatrick, and J. J. McAuley, "Deep performer: Score-to-audio music performance synthesis," in *Proc. IEEE Int. Conf. Acoust., Speech, Signal Process.*, Singapore, 2022 pp. 951–955.
- [29] Y. Wu et al., "MIDI-DDSP: Detailed control of musical performance via hierarchical modeling," in Proc. Int. Conf. Learn. Representations, 2022.
- [30] H. Kim, S. Choi, and J. Nam, "Expressive acoustic guitar sound synthesis with an instrument-specific input representation and diffusion outpainting," in *Proc. 2024 IEEE Int. Conf. Acoust., Speech Signal Process.*, 2024, pp. 7620–7624.
- [31] A. van den Oord et al., "WaveNet: A generative model for raw audio," in Proc. ISCA Speech Synth. Workshop, Sunnyvale, USA, 2016.
- [32] J. H. Engel et al., "Neural audio synthesis of musical notes with wavenet autoencoders," in *Proc. Int. Conf. Mach. Learn.*, Sydney, Australia, 2017, vol. 70, pp. 1068–1077.
- [33] A. Agostinelli et al., "MusicLM: Generating music from text," 2023, arXiv:2301.11325.
- [34] F. Schneider, Z. Jin, and B. Schölkopf, "Moûsai: Text-to-music generation with long-context latent diffusion," 2023, arXiv:2301.11757.
- [35] N. Zeghidour, A. Luebs, A. Omran, J. Skoglund, and M. Tagliasacchi, "Soundstream: An end-to-end neural audio codec," *IEEE/ACM Trans. Audio, Speech Lang. Process.*, vol. 30, pp. 495–507, 2022.
- [36] X. Riley, D. Edwards, and S. Dixon, "High resolution guitar transcription via domain adaptation," in *Proc. IEEE Int. Conf. Acoust., Speech, Signal Process.*, Seoul, South Korea, 2024, pp. 1051–1055.
- [37] A. M. Turing, "I.—Computing Machinery and Intelligence," *Mind*, vol. LIX, no. 236, pp. 433–460, 1950.
- [38] International Telecommunications Union, "ITU-R Rec. BS.1534-3: Method for the subjective assessment of intermediate quality levels of coding systems," 2015.
- [39] D. P. Kingma and J. Ba, "Adam: A method for stochastic optimization," in *Proc. Int. Conf. Learn. Representations*, San Diego, California, USA, 2015.
- [40] A. Q. Nichol and P. Dhariwal, "Improved denoising diffusion probabilistic models," in Proc. 38th Int. Conf. Mach. Learn., Jul. 2021, pp. 8162–8171.

- [41] J. Song, C. Meng, and S. Ermon, "Denoising diffusion implicit models," in Proc. 9th Int. Conf. Learn. Representations, Austria, May 2021.
- [42] J. Ho and T. Salimans, "Classifier-free diffusion guidance," *NeurIPS Workshop Deep Generative Models Downstream Appl.*, 2021.
- [43] M. Schoeffler et al., "webMUSHRA—A comprehensive framework for web-based listening tests," J. Open Res. Softw., vol. 6, no. 1, 2018, Art. no. 8.
- [44] M. Heusel, H. Ramsauer, T. Unterthiner, B. Nessler, and S. Hochreiter, "GANs trained by a two time-scale update rule converge to a local nash equilibrium," in *Proc. Adv. Neural Inf. Process. Syst. 30: Annu. Conf. Neural Inf. Process. Syst.*, Long Beach, CA, USA, 2017, pp. 6626–6637.
- [45] J. Shor et al., "Towards learning a universal non-semantic representation of speech," in *Proc. Annu. Conf. Int. Speech Commun. Assoc.*, Shanghai, China, 2020, pp. 140–144.
- [46] S. Hershey et al., "CNN architectures for large-scale audio classification," in *Proc. IEEE Int. Conf. Acoust., Speech, Signal Process.*, New Orleans, Louisiana, USA, Mar. 2017, pp. 131–135.
- [47] L. van der Maaten and G. Hinton, "Visualizing data using t-sne," J. Mach. Learn. Res., vol. 9, no. 86, pp. 2579–2605, 2008.



Ben Maman received the B.Sc. degree in mathematics and computer science, and the M.Sc. degree in mathematics from the Hebrew University, Jerusalem, Israel, in 2017 and 2019, respectively. He pursued his Ph.D. degree at the Blavatnik School of Computer Science in Tel-Aviv University under the supervision of Prof. Amit H. Bermano (Ph.D. dissertation currently under review). In 2024, he joined the International Audio Laboratories, Erlangen, Germany, as a Researcher in the group of Prof. Meinard Müller. His main research interests include diffusion-based music

synthesis, automatic music transcription, and human-computer interaction.



Johannes Zeitler received the B.Sc. degree in eletrical engineering and the M.Sc. degree in signal processing and communications engineering from Friedrich-Alexander-Universität Erlangen-Nürnberg, Erlangen, Germany, in 2019 and 2021, respectively. In 2022, he joined the International Audio Laboratories Erlangen, Germany, where he is currently working toward the Ph.D. degree, under the supervision of Prof. Meinard Müller. His research focuses on alignment techniques in music processing.



**Meinard Müller** received the Diploma in mathematics and the Ph.D. degree in computer science from the University of Bonn, Bonn, Germany, in 1997 and 2001, respectively, and the Habilitation degree in multimedia retrieval in Bonn, Germany, in 2007. He conducted postdoctoral studies in Japan in 2003. From 2007 to 2012, he was a Senior Researcher with Saarland University, Saarbrücken, Germany, and was also affiliated with the Max Planck Institute for Informatics, Saarbrücken. Since 2012, he has held the Professorship for Semantic Audio Signal Process-

ing with the International Audio Laboratories Erlangen, a joint institution of Friedrich-Alexander-Universität Erlangen-Nürnberg and the Fraunhofer Institute for Integrated Circuits IIS. His research interests include music processing, music information retrieval, audio signal processing, and motion processing. From 2010 to 2015, he was a member of the IEEE Audio and Acoustic Signal Processing Technical Committee, was a member of the Senior Editorial Board of the *IEEE Signal Processing Magazine* from 2018 to 2022, and served on the Board of Directors of the International Society for Music Information Retrieval from 2009 to 2021, acting as its president from 2020 to 2021. In 2020, he was elevated to IEEE Fellow for his contributions to music signal processing.



Amit H. Bermano received the bachelor's and master's degrees from the The Technion—Israel Institute of Technology, Haifa, Israel, under the supervision of Prof. Craig Gotsman. He has conducted his Ph.D. studies at ETH Zurich under the supervision of Prof. Dr. Markus Gross, in collaboration with the Computational Materials Group of Disney Research Zurich. He is currently an Associate Professor with the Blavatnik School of Computer Science, Tel-Aviv University, Tel Aviv, Israel. He was a Postdoctoral Researcher with the Princeton Graphics Group, hosted by Prof.

Szymon Rusinkiewicz and Prof. Thomas Funkhouser. Before that, he was a Postdoctoral Researcher with Disney Research Zurich. His research interests include computer graphics, computer vision, computational fabrication (3D printing), and music information retrieval, using tools of geometry processing, machine learning, and generative artificial intelligence.